Author’s response to reviews

Title: CLINICAL SPECTRUM AND FACTORS IMPACTING OUTCOME OF CANDIDA AURIS: A SINGLE CENTER STUDY FROM PAKISTAN

Authors:
Muneeba Ahsan Sayeed (muneeba.sayeed@yahoo.com)
Joveria Farooqi (joveria.farooqi@aku.edu)
Kausar Jabeen (kausar.jabeen@aku.edu)
Safia Awan (safia.awan@aku.edu)
Syed Faisal Mahmood (faisal.mahmood@aku.edu)

Version: 2 Date: 13 Apr 2019

Author’s response to reviews:

Ref: Re-submission of manuscript titled “CLINICAL SPECTRUM AND FACTORS IMPACTING OUTCOME OF CANDIDA AURIS: A SINGLE CENTER STUDY FROM PAKISTAN”

Dear Editors,

We are resubmitting the above titled manuscript after carrying out the minor changes suggested by the Reviewer 2. We have responded to each of the comments and mentioned the line numbers. Track changes and highlights have been removed.
Thank you very much for your consideration.

MINOR CHANGES:
Silke Schelenz (Reviewer 2): The authors have addressed most of the issues raised and revised the manuscript well. There are some minor matters that could be improved:

Line 136 It might be better to refer to the voriconazole TDM in the past tense: ...was not available.

Done. Methods section, line 136, page 7

Line 164: I still sting the authors should state that the rest of the yeast isolates were 'presumed to be C auris'. This is because of the formally unconfirmed ID using MALDI or sequencing at the time of publication.